Author's response to reviews

Title: Achondroplastic girl manifesting enchondromatosis and ossification of the spinal ligaments

Authors:

Ali Al Kaissi (ali.alkaissi@osteologie.at)
Klaus Klaushofer (klaus.klaushofer@osteologie.at)
Monika Rumpler (monika.rumpler@osteologie.at)
Franz Grill (franz.grill@oss.at)

Version: 6 Date: 5 March 2008

Author's response to reviews: see over
Dear Editor,

The paper was modified in accordance with the reviewers’ comments. However, we attached one MRI Image instead of 3. We clarified the importance of MRI images in studying patients with spinal ankylosis in our response to the reviewers.

Sincerely yours
Al Kaissi